Author's response to reviews

Title: Exploring Associations Between Multipollutant Day Types And Asthma Morbidity: Epidemiologic Applications Of Self-Organizing Map Ambient Air Quality Classifications

Authors:

John L Pearce (pearcejo@musc.edu)
Lance A Waller (lwaller@emory.edu)
James A Mulholland (james.mullholland@ce.gatech.edu)
Stefanie E Sarnat (sebelt@emory.edu)
Matthew J Strickland (mistric@emory.edu)
Howard H Chang (howard.chang@emory.edu)
Paige E Tolbert (ptolber@emory.edu)

Version: 2
Date: 12 February 2015

Author's response to reviews:

- Included a Title page as page 1. The title page should include the title, authors, author affiliations, email addresses and corresponding author details.

- Included the Abstract on page 2.